Author’s response to reviews

Title: An oral health literacy intervention for Indigenous adults in a rural setting in Australia

Authors:

Eleanor J Parker (eleanor.parker@adelaide.edu.au)
Gary Misan (gary.misan@unisa.edu.au)
Alwin Chong (alwin.chong@menzies.edu.au)
Helen Mills (helen.m.mills@unisa.edu.au)
Kaye Roberts-Thomson (kaye.robertsthomson@adelaide.edu.au)
Alice M Horowitz (ahorowit@umd.edu)
Lisa M Jamieson (lisa.jamieson@adelaide.edu.au)

Version: 2 Date: 8 March 2012

Author’s response to reviews:

Dear Editor

An updated manuscript has been submitted, after noting 3 editing errors.

Copies of requested documents have been emails (evidence of ethical approval funding).

Yours sincerely

Eleanor Parker